Author's response to reviews

Title: Do general practitioners adhere to the guideline on infectious conjunctivitis? Results of the Second Dutch National Survey of General Practice

Authors:

Remco P Rietveld (r.p.rietveld@amc.uva.nl)
Gerben ter Riet (g.terriet@amc.uva.nl)
Patrick JE Bindels (p.j.bindels@amc.uva.nl)
Francois G Schellevis (francois.schellevis@nivel.nl)
Henk CPM van Weert (h.c.vanweert@amc.uva.nl)

Version: 3 Date: 10 September 2007

Author's response to reviews: see over
10 September 2007

Dear Sir, Madam,

Attached please find the revised manuscript entitled “Do general practitioners adhere to the guideline on infectious conjunctivitis? Results of the Second Dutch National Survey of General Practice” by Remco P Rietveld, Gerben ter Riet, Patrick JE Bindels, François G Schellevis, Henk CPM van Weert

The article has been extensively edited. Furthermore, we have formatted all changes as suggested by the Biomed Central Editorial Production Team in their email dated August 6, 2007.

We hope that our paper is now ready for publications after this revision.

Yours sincerely,
Also on behalf of my co-authors,

Remco P. Rietveld, General Practitioner